Author's response to reviews

Title: Alcohol, betel-nut and cigarette consumption are negatively associated with health promoting behaviors in Taiwan: A cross-sectional study

Authors:

Su-Er Guo (seguo@gw.cgust.edu.tw)
Tung-Jung Huang (donaldhuang@adm.cgmh.org.tw)
Jui-Chu Huang (huangrz1204@gmail.com)
Ming-Shyan Lin (mingshyan@gmail.com)
Rei-Mei Hong (rmhong@gw.cgust.edu.tw)
Chia-Hao Chang (chchang@gw.cgust.edu.tw)
Mei-Yen Chen (meiyen@gw.cgust.edu.tw)

Version: 2 Date: 12 September 2012

Author's response to reviews: see over
September 12, 2012

BMC Public Health

Dear Dr. Jimmar Dizon,

I am submitting the revised article entitled “Alcohol, betel-nut and cigarette consumption are negatively associated with health promoting behaviors in Taiwan: A cross-sectional study” (9623051277851729) incorporating the requested revisions described in the email correspondence.

My co-authors and I first want to sincerely thank you for the useful suggestions. We feel that our paper has benefited substantially from your suggestions and that the revised manuscript is a significant improvement over the previous version.

We have made some changes according to your suggestions. Please see the response to suggestion letter (Attached on next page).

Again, we sincerely thank you for your support and guidance.

Looking forward to hearing from you soon.

Sincerely,

Su-Er

Su-Er Guo, RN, PhD
Associate Professor and Director of the Chronic Diseases and Health Promotion Research Center, CGUST, Taiwan
seguo@gw.cgust.edu.tw
Manuscript number: 9623051277851729

Thank you for giving us support and feedback on our manuscript. We have followed the editor’s comments. Please see the changes in red font in the manuscript.

1. Regarding the need to provide the specific name of the Ethics committee

   Ans. Please see the red font on p8.

2. Regarding the manuscript is suboptimally organized and the sampling plan, selection of participants, and interview procedure need more detail in order to be evaluable scientifically....

   Ans. We have revised these points in red font on page 7-9.